Author's response to reviews

Title: Painful Ophthalmoplegia with Normal Cranial Imaging

Authors:

Chih-Hsien Hung (jasonhung0701@gmail.com)
Kuo-Hsuan Chang (gophys5128@adm.cgmh.org.tw)
Chun-Che Chu (1939chu@adm.cgmh.org.tw)
Ming-Feng Liao (mingfengliag@hotmail.com)
Hong-Shiu Chang (hachang@adm.cgmh.org.tw)
Rong-Kuo Lyu (lyu5172@adm.cgmh.org.tw)
Yi-Ming Wu (jimmy_CGMH@hotmail.com)
Yao-Liang Chen (8828cyl@gmail.com)
Chiou-Lian Lai (cllai@cc.kmu.edu.tw)
Hsian-Jung Tseng (allebjht@gmail.com)
Long-Sun Ro (cgrols@adm.cgmh.org.tw)

Version: 3 Date: 3 January 2014

Author's response to reviews: see over
Dear Editor:

Thank you for the editorial consideration of the manuscript MS: 2009726171033426 “Painful Ophthalmoplegia with Normal Cranial Imaging”. We deeply appreciated the reviewers’ agreement on our manuscript. We added a section of acknowledgements at the end of manuscript. We also revised our manuscript according to the reviewer’s comment. The points raised by the reviewers were addressed as below. Your editorial consideration of the manuscript is highly appreciated.

Reviewer: Ping-Kun Chen
Minor Essential Revisions
1. Page 13, line 6-8. Please re-check the glucocorticoids treatment in "episode of attacks" or "number of patients". 24 cases, 88.9% seems mean 24 episodes of 27 attacks. However, 13 cases, 54.2% seems mean 13 of 25 patients.
Reply to reviewer:
(Page 13 and line 6) Thank reviewer for careful check-up. We have revised the sentence to “Glucocorticoids were prescribed in 24 episodes (88.9%) of benign THS and 4 of OM (80.0%). Therapeutic trial of glucocorticoids was used in 13 episodes (50.0%) of ocular diabetic neuropathy because of initially inconclusive diagnoses.”

2. Page 17, line 17, should be ICHD-2 criteria
Reply to reviewer:
Thank your for pointing it out. We have corrected it in the revised manuscript.

Sincerely yours,
Chih-Hsien Hung, MD; Long-Sun Ro, MD, PhD